Author’s response to reviews

Title: Case Report: Histopathology of a Retrocorneal Membrane after Descemet Membrane Endothelial Keratoplasty

Authors:
Tarek Bayyoud (tarek_bayyoud@gmx.de)
Jens Rohrbach (martin.rohrbach@med.uni-tuebingen.de)
Karl Bartz-Schmidt (u.bartz-schmidt@med.uni-tuebingen.de)
Sebastian Thaler (sebastian.thaler@med.uni-tuebingen.de)

Version: 2 Date: 14 Dec 2018

Author’s response to reviews:
Response to Reviewer/Reviewers:

Point-by-point response with highlighted changes to the manuscript:

Editor’s comments:
- The title of the case report was changed to meet the CARE guidelines; the words “case reports” were added to the title of the article.

We apologize for our error and thank you for your conscientiousness.